Author's response to reviews

Title: Neurophysiologic effects of spinal manipulation in patients with chronic low back pain

Authors:

Brian C Clark (clarkb2@ohio.edu)
David A Goss (dg180809@ohio.edu)
Stevan Walkowski (walkowski@oucom.ohiou.edu)
Richard L Hoffman (hoffman@oucom.ohiou.edu)
Andrew Ross (aross.635@gmail.com)
James S Thomas (thomasj5@ohio.edu)

Version: 2 Date: 4 July 2011

Author's response to reviews: see over
March 30, 2011

Re: Manuscript Submission

Dear BMC Musculoskeletal Disorders,

Please accept our manuscript titled “Neurophysiologic effects of spinal manipulation in patients with chronic low back pain” for consideration for publication in BMC Musculoskeletal Disorders.

Low back pain is the most common, disabling and costly musculoskeletal disorder in the western world, and manual therapies (e.g., spinal manipulation) is one of the most common treatment interventions used for low back pain. However, to date there is very little scientific investigation into the mechanisms of manual therapies in low back pain. Accordingly, this work utilizes innovative techniques to discover novel findings as it relates to the effects and consequences of spinal manipulation. For these reasons we feel this manuscript is an outstanding fit for BMC Musculoskeletal Disorders.

Please let me know if you have any questions.

Sincerely,

Brian C. Clark, Ph.D.
Executive Director, Ohio Musculoskeletal & Neurological Institute (OMNI)
Associate Professor of Physiology, Department of Biomedical Sciences
Ohio University Heritage College of Medicine